Author's response to reviews

Title: Interventions Aimed at Increasing Research Use in Nursing: A Systematic Review

Authors:

David S Thompson (dst3@ualberta.ca)
Carole A Estabrooks (carole.estabrooks@ualberta.ca)
Shannon Scott-Findlay (shannon.scott-findlay@ualberta.ca)
Katherine Moore (katherine.moore@ualberta.ca)
Lars Wallin (lars.wallin@karolinska.se)

Version: 5 Date: 30 April 2007

Author's response to reviews:

Dear Dr Eccles,
As requested, I have reviewed the changes made by your copyeditor. I am happy with all changes and accepted them in the manuscript. Attached you will find a 'clean copy' of our manuscript entitled 'Interventions aimed at increasing research use in nursing: a systematic review'.

Best wishes,

David Thompson